# Investigating the micro-level dynamics of water reuse adoption by farmers and the impacts on local water resources using an agent-based model

Farshid Shoushtarian<sup>1</sup>, Masoud Negahban-Azar<sup>1\*</sup>, and Andrew Crooks<sup>2</sup> <sup>1</sup>Department of Environmental Science and Technology, University of Maryland, USA <sup>2</sup>Department of Geography, University at Buffalo, USA

#### Abstract

Agricultural water reuse is gaining momentum to address freshwater scarcity worldwide. The main objective of this paper was to investigate the micro-level dynamics of water reuse adoption by farmers at the watershed scale. An agent-based model was developed to simulate agricultural water consumption and socio-hydrological dynamics. Using a case study in California, the developed model was tested, and the results showed that agricultural water reuse adoption by farmers is a gradual and time-consuming process. In addition, results also showed that agricultural water reuse could significantly decrease the water shortage (by 57.7%) and groundwater withdrawal (by 74.1%). Furthermore, our results suggest that recycled water price was the most influential factor in total recycled water consumption by farmers. Results also showed how possible freshwater shortage or groundwater withdrawal regulations could increase recycled water use by farmers. The developed model can significantly help assess how the current water reuse management practices and strategies would affect the sustainability of agricultural water resources.

#### Keywords

Water reuse; agent-based modelling; agricultural water management; recycled water for irrigation

#### Code availability

The WRAF (water reuse adoption by farmers) model presented in this paper and its complete description following the Overview, Design concepts, Details, and Decision-making (ODD) (Grimm et al., 2006) protocol can be found at https://www.comses.net/codebase-release/cc6d551e-cf0f-472e-a54b-28591cd39b4d/.

# 1. Introduction

Agriculture is the largest water-consuming sector worldwide, responsible for almost 70% of the world's total freshwater consumption (Suri et al., 2019). In the US, for example, in 2018, 231,474 farms (226,219 million m<sup>2</sup>) were irrigated with 102,872 million m<sup>3</sup> of water (USDA-NASS, 2019). During this time frame, the primary sources of irrigation in the US were as follows: 1) groundwater from on-farm wells (146,496 million m<sup>2</sup> of land irrigated with 49.8% of the total irrigation water consumption); 2) on-farm surface water 25,495 million m<sup>2</sup> of land irrigated with 10% of the total irrigation water consumption); and 3) off-farm water from a variety of sources (64,345 million m<sup>2</sup> of land irrigated with 40.2% of the total irrigation water consumption) (USDA-NASS, 2019).

Correspondence: Contact M. Negahban-Azar at <u>mnazar@umd.edu</u> Cite this article as: Shoushtarian, F., Negahban-Azar, M., & Crooks, A. Investigating the micro-level dynamics of water reuse adoption by farmers and the impacts on local water resources using an agent-based model Socio-Environmental Systems Modelling, vol. 4, 18148, 2022, doi:10.18174/sesmo.18148

This work is **licensed** under a <u>Creative Commons Attribution-NonCommercial 4.0 International</u> <u>License</u>.





As such, the agriculture sector is highly dependent on water availability (Mendelsohn & Dinar, 2003), and any water shortage diminishes the number of productive farms, irrigated areas, and crops, thus impacting food production. Therefore, access to adequate water resources is crucial to the future of global agriculture, food security, and the economy (Paul et al., 2020).

Solutions for addressing water scarcity can generally be categorized into two major groups: increasing water supply and decreasing demand. While there are several ways to increase water supply (e.g., water reuse, desalination, and water transfer), agricultural water reuse is one of the most prominent ones by introducing a reliable alternative water resource (Paul et al., 2020). Using treated wastewater for irrigation is the dominant water reuse application globally (Eslamian, 2016). Agricultural water reuse provides various benefits, such as reducing freshwater withdrawal, managing/recovering nutrients, decreasing pollutants discharge, avoiding groundwater pollution, closing the water cycle, and increasing water supply reliability (Shoushtarian & Negahban-Azar, 2020).

Although agricultural water reuse has excellent potential to alleviate global water scarcity, its challenges make it less likely to be chosen by water resources decision-makers than other alternative water supplies. Agricultural water reuse includes various challenges such as water quality concerns, social acceptance, technical feasibility, socio-economic factors, regulatory considerations, and potential conflicts between stakeholders (Shoushtarian & Negahban-Azar, 2020). Generally, agricultural water reuse challenges can be categorized into seven categories: human health, environmental health, technical, social, legal, and socio-economic (Shoushtarian & Negahban-Azar, 2020). Therefore, it is necessary for water resources decision-makers to consider multiple agricultural water reuse challenges to plan and manage safe agricultural water reuse practices worldwide.

Planning and managing agricultural water reuse projects without paying adequate attention to different water system components, their complex interactions, and the exogenous factors affecting them can result in various ramifications. Socio-economic factors are among the most critical factors for successful agricultural water reuse projects, potentially turning the project into a failure if not appropriately addressed (Lazarova & Bahri, 2005; Asano et al., 2007; Sheikh et al., 2018). It is also necessary to investigate the dynamics of coupled human-environment systems in agricultural water reuse projects. The micro-scale dynamics of these projects are of paramount importance for decision-makers to successfully identify and implement best management practices. For example, policies set at the local level (e.g., irrigation district) can alter the micro-dynamics of agricultural water reuse adoption, resulting in changing the macro-scale dynamics of water resources systems at the watershed scale (e.g., groundwater over-drafting caused by the increase in using groundwater).

Agent-based modeling (ABM) is one of the methods which has been used to study complex systems (e.g., coupled human-environment systems) (Elsawah et al., 2015; Filatova et al., 2013; Huber et al., 2018; Janssen and Ostrom, 2006). Although this method is relatively new, it is becoming a prominent way to analyze, model, and simulate socio-hydrological complex systems in recent years (Akhbari and Grigg, 2013). Comparing agent-based and other modeling paradigms, the advantages of this method are as follows: 1) this method uses a bottom-up approach which enables us to capture the emergent phenomena in the system being studied; 2) it is an efficient way to study the socio-economic and socio-demographic factors geospatially using its natural environment; 3) it provides the opportunity for including elements of randomness in models; and 4) it develops a platform to create agents that are autonomous, adaptive, and have unlimited numbers of impacting parameters and rules (Bert et al., 2015; Bithell and Brasington, 2009; Crooks et al., 2018; Groeneveld et al., 2017; Crooks and Castle 2012; Kelly et al., 2013).

Various agent-based models have been developed to study water resources management and sociohydrological systems. For example, Nikolic et al. (2015) integrated ABM with three other analytical tools (geographic information system, system dynamics, and hydrologic simulation) to develop a decision support system for integrated water resources management of the Upper Thames River watershed (Ontario, Canada). In another study, Tillman et al. (2001) explored municipal water supply stakeholders' decision-making processes to test various management scenarios for developing adaptive planning and management strategies using ABM. Wise and Crooks (2012) studied local social and institutional structures in the physical water systems in northern New Mexico to investigate traditional farming sustainability. Rasoulkhani et al. (2018) developed an agent-based model to explore the dynamics in water conservation technology adoption of residential users, using Miami Beach, Florida, as the case study. In another example, Kandiah et al. (2019a) used an ABM framework to study urban water reuse adoption by consumers and infrastructure expansion (the Town of Cary, North Carolina). Moreover, Kock (2008) developed a socio-hydrological agent-based model to study how increasing water resources management institutional capacity would decrease conflict levels in the USA and Spain. In another study, Pouladi et al. (2019) proposed a socio-hydrological ABM framework to study the performance of complex water resources systems to restore Urmia Lake (Iran). A socio-hydrological ABM framework was developed by Mashhadi Ali et al. (2017) for simulating urban water supply and demand with different climate change scenarios, studying Raleigh, North Carolina, as a case study. In another example, Farhadi et al. (2016) developed a socio-hydrological ABM framework to find the best policy mechanisms for allocating groundwater to users in Daryan Aquifer, Fars Province, Iran. Pope and Gimblett (2015) used a coupled model (agent-based and Bayesian) to simulate the complex interactions of decision-makers (water demand) and environmental conditions in the Rio Sonora watershed, Mexico. A socio-hydrological ABM framework was created by Akhbari and Grigg (2013) to simulate the conflicts between parties in the San Joaquin (California) watershed and find the best solutions scenarios for water resources management in this watershed.

The development of agent-based models for water resources management can help to improve understanding on how how water demand and supply interact with the hydrologic cycle over time and space. For instance, agent-based models can help determine water conservation's main incentives in a watershed by simulating different water users' responses (i.e., urban users) under different management scenarios (Rasoulkhani et al., 2018). However, it is evident that as the main step toward further development of such models, we need to gain an understanding of how the current water management operates in the watersheds and how stakeholders may impact the system. For example, it is necessary to capture the complex and adaptive dynamics of sociohydrological systems inherent in sustainable water resources management when alternative water sources (e.g., recycled water) are introduced at the watershed scale (Kandiah et al., 2019b). We would argue that this is of paramount importance for water resource managers, especially when it comes to managing agricultural water reuse projects sustainably due to the complexities involved in these projects. However, to the best of our knowledge, very limited studies have investigated the adaptive and complex dynamics of socio-hydrological systems inherent in agricultural water reuse projects, especially regarding capturing the dynamics of agricultural water reuse adoption by farmers and its impacts on local water resources. There has also been very limited research on quantifying the potentials of agricultural water reuse projects in addressing water shortage and groundwater over-drafting problems so far.

Therefore, this study attempts to fill these gaps by investigating the micro-level dynamics of agricultural water reuse adoption by farmers and its impacts on local water resources. It further explores how agricultural water reuse adoption, as a community-wide behavior, emerges from interactions, relationships, and dependencies between farmers and the local water resources, as the water supply system shifts from having only conventional water sources to a mix of conventional and alternative water sources. This study further explores the most critical parameters (e.g., the unit price of recycled water) when it comes to the sustainability of agricultural water reuse projects and local water resources to beneficially help water resource managers make better-informed decisions for managing these projects. In summary, this study's main objective is to fill this gap in the literature by developing an agent-based model for simulating the socio-hydrologic dynamics of agricultural water reuse projects to identify the best planning and management practices for these projects. In the remainder of this paper, Section 2 introduces our study area and methodology, Section 3 presents our results, Section 4 discusses the results of this study and its limitations, and Section 5 summarizes the paper and suggests areas for further study.

# 2. Methodology

#### 2.1 Study area

The Del Puerto Water District (DPWD) in Central Valley, California, was selected as the study area (Figure 1). DPWD provides irrigation water for approximately 182.1 million m<sup>2</sup> of agricultural land in three counties: Stanislaus, San Joaquin, and Merced (RMC Water and Environment, 2013). DPWD's main water supply is provided through a contract with the United States Bureau of Reclamation (USBR). It delivers 172,946,231 m<sup>3</sup> per year of water from the Central Valley Project (CVP) (RMC Water and Environment, 2013). However, due to recent drought conditions and limitations on pumping from the San Joaquin Bay Delta, CVP allocation has not been completely provided for the DPWD (RMC Water and Environment, 2013). Since 1983, fallowing practices

have increased in this area as alternative water resources in DPWD (e.g., groundwater and temporary transferred water) have been unreliable, unsustainable, and unaffordable (RMC Water and Environment, 2013). California State Water Resources Control Board adopted its Recycled Water Policy on January 22, 2013, and shortly after this, the North Valley Regional Recycled Water Program (NVRRWP) was started to address the DPWD's water shortage problem (RMC Water and Environment, 2013).



Figure 1: Map of this research's case study area (Del Puerto Water District, CA, USA)

In the NVRRWP, two treatment plants, one in Modesto and another in Turlock, are responsible for treated wastewater. These plants were planned to transfer the wastewater to the DPWD to supplement its water resources for agricultural irrigation in the region (RMC Water and Environment, 2013). Based on the NVRRWP feasibility study (RMC Water and Environment, 2013), the DPWD average water demand in the future (2045) will be 0.1 billion m<sup>3</sup> per year, while the expected future average supply (precipitation and CVP) will be between 34,537,440 and 60,440,520 m<sup>3</sup> per year. Therefore, DPWD is expected to have a water shortage of 49,339,200 to 74,008,800 m<sup>3</sup> per year. Furthermore, the feasibility study (RMC Water and Environment, 2013) also interviewed 12 farmers to assess their perceptions toward water reuse. Overall, these interviews showed that

the farmers did not have an opposing perspective toward agricultural water reuse. However, the farmer's main concern was whether the DPWD could deliver recycled water at an affordable cost.

To successfully simulate the study area and explore water reuse adoption, farms, their characteristics (e.g., crop type, area, and water requirement), and existing infrastructure that provides the recycled water were needed. Therefore, the National Agricultural Statistics Service Cropland Data Layers (NASS-CDL) were used to identify and collect the study area land use and land cover data. This data were used to determine each farm's crop. The California State Geoportal website was also used to find and gather the farms' shapefiles to calculate their area. The United States Department of Agriculture (USDA) census of agriculture in 2017 and the NVRRWP feasibility study were used to determine crop's daily irrigation requirements. These data were needed for each farm to determine its water requirement (discussed further in Section 2.2.1).

Moreover, data on farmers' opinions on the use of recycled water, the location of wastewater treatment plants in the study area, and their discharge flow rate were needed. Survey results from Suri et al. (2019) were used to determine farmers' perceptions toward using recycled water, which will be described more in the following sections. It should be noted that the survey was conducted by Suri et al. (2019); we used the results of this study for developing the model presented in this paper (which we will detail more in Section 2.2). The US Environmental Protection Agency's (EPA) Enforcement and Compliance History Online database were also utilized to identify and collect the existing wastewater treatment plants' location and discharge flow rate (Table 1).

Data	Source	Link
Land use and land cover data	National Agricultural Statistics Service Cropland Data Layers (NASS- CDL) with 30m resolution	https://nassgeodata.gmu.edu/CropScape/
Shapefiles of agricultural parcels	California State Geoportal	https://gis.data.ca.gov/
Survey data from farmers	U.S. farmers' opinions on the use of nontraditional water sources for agricultural activities (Suri et al., 2019).	Raw data were obtained from the authors.
Crop's irrigation requirements	1) The United States Department of Agriculture (USDA) census of	1) https://www.nass.usda.gov/Publications/ AgCensus/2017/index.php
	agriculture (2017), 2) the NVRRWP feasibility study	2) RMC Water and Environment (2013)
Wastewater treatment plants' location and discharge flow rate	The United States Environmental Protection Agency's (EPA) Enforcement and Compliance History Online	https://echo.epa.gov/facilities/facility- search/results

 Table 1: Input data used for developing the agent-based model in this research.

#### 2.2 ABM Framework

This section describes the agent-based model developed for simulating the dynamics of agricultural water reuse adoption by farmers (WRAF) and its effects on water resources in the study area (Figure 2a). It was developed as an exploratory tool for scenario analysis. The WRAF model simulates a virtual agricultural area where several autonomous farms operate. It also simulates these farms' water consumption dynamics. The developed model includes two types of agents: farmers and wastewater treatment plants. In general, farmer agents are the main water-consuming agents, and wastewater treatment plant agents are recycled water providers in the WRAF model. Dynamic simulation of agricultural water supply and demand in the area allows the user to observe the results of various irrigation water management scenarios, including the use of recycled water. The model also enables the user to apply multiple climate change scenarios, including normal, moderate drought, severe drought, and wet weather conditions. The model was developed using NetLogo 6.1.1 (Wilensky, 1999). The following sections describe the model structure and explain the agents in the model in more detail.

### 2.2.1 Agents

#### 2.2.1.1 Farmers

Almond is the most dominant crop in the study area (Table 2) (RMC Water and Environment, 2013). Compared to other cultivated crops in the area, almond farms require most of the water supply for irrigation (RMC Water and Environment, 2013). The fact that almond growing requires a significant amount of water makes it vulnerable to water shortages. Almond trees stay in production for 25 years or more and require a constant water source during their lifetime. Furthermore, starting an almond orchard requires significant capital investment and is considered a high-value crop, producing high profits for farmers compared to other crops (RMC Water and Environment, 2013). Considering all these factors and the need to keep the model as simple as possible for the sake of parsimony, we only included the almond farmers as farmer agents (however, as we will discuss in Section 4, this could be an area of further work).

**Table 2:** Total irrigation demand and area under cultivation of crops grown in the study area (RMC Water and Environment,2013)

Сгор	Total irrigation demand (m <sup>3</sup> /year) [%]	Area (Hectare) [%]
Almonds	6,663,382 [44.14%]	5,740 [43.61%]
Apricots	3,219,259 [21.33%]	1,019 [7.75%]
Other deciduous fruits and nuts	2,521,603 [16.70%]	480 [3.64%]
Cherries	970,749 [6.43%]	210 [1.59%]
Beans	632,775 [4.19%]	899 [6.83%]
Vineyard	593,057 [3.93%]	150 [1.14%]
Tomatoes	451,084 [2.99%]	1,575[11.97%]
Walnuts	38,855 [0.26%]	721 [5.48%]
Oats/barley	5,057 [0.03%]	1,000 [7.60%]
Alfalfa/mixed pasture	0.00 [0.00%]	672 [5.10%]
Vegetables	0.00 [0.00%]	370 [2.81%]
Melons	0.00 [0.00%]	161 [1.22%]
Grapefruit/lemons/oranges	0.00 [0.00%]	138 [1.05%]
Flowers, nursery	0.00 [0.00%]	27 [0.20%]
Total	15,095,822 [100.00%]	13,161 [100.00%]

Each almond farm is represented as an individual agent (i.e., farmer) in the model. Farm characteristics, including the farm area and crop type, were added to the model based on the available data (Table 1). It was assumed that farmers would irrigate their farms from available water resources during the irrigation season using the following assumptions (Figure 2b). First, based on equations (1) and (2), farmers calculate gross crop irrigation requirement (GCIR) (FAO, 2015) to determine the amount of water that they need for irrigation. Second, farmers determine if they have a water shortage based on a water mass balance, subtracting GCIR from their primary water supply shares. Those farmers with no water shortage continue their farming as usual. However, farmers with water shortages will look for alternative water sources to irrigate their farms. Then among the available alternative water resources, farmers will choose the cheapest one. The assumption is that farmers would buy the amount of water they need for irrigation if they can afford it.

This process repeats during the irrigation season daily. At the end of the farming season, farmers sell their crops (discussed further below). Alternative water sources included in the WRAF model were groundwater, transferred water, and recycled water (from wastewater treatment plants). Of note is that farmers take recycled water into account as an alternative water resource after adopting water reuse (accepting to use recycled water for irrigation). Water reuse adoption is determined using a sub-model, explained in the following.

$$ET_c = K_c \times ET_0 \tag{1}$$

Gross Crop Irrigation Requirement (GCIR) = 
$$\frac{ET_c - EP}{IE}$$
 (2)

Where  $K_c$  is the crop coefficient,  $ET_0$  is the evapotranspiration for the reference crop (mm/day),  $ET_c$  is the crop evapotranspiration (mm/day), EP is the effective precipitation (mm/day), and IE is the irrigation efficiency (FAO, 2015).



Figure 2: (a) WRAF framework; (b) Farmers' decision-making flowchart

There were 244 almond farmers in DPWD in 2018. Among these farms, 19%, 38%, and 43% of the almond farmers use sprinkler (irrigation efficiency (IE) = 80%), surface irrigation (IE = 75%), and drip irrigation (IE = 85%) technology, respectively (RMC Water and Environment, 2013). Farmers calculate their GCIR based on Table 3. It should be mentioned that daily GCIRs were considered constant during each month. So, farmers calculate their daily GCIRs by dividing monthly GCIRs (Table 3) by days of the month (28, 30, or 31 days).

USBR provided farmers with a maximum of 0.9 m<sup>3</sup>/m<sup>2</sup> in each year for all the DPWD area. For scenario analysis, the user can change the available USBR water supply percentage between 0 and 100 percent. The starting age of almond trees for each farm was chosen randomly (33% of farms were between 0-8 years of age; 33% of farms were between 9-16 years of age; 34% of farms were between 17-25 years of age). It often takes three years from the day almond trees are planted till they can be harvested. In this model, almond farmers could sell their almonds from year three until year 25 based on equation (3). It was also assumed that another set of almond trees would again be planted after year 25 was finished.

Month	30-year EP, 80% of average (mm)	ET, Sprinkler (mm)	ET, Surface (mm)	ET, Drip (mm)
Jan.	51.82	43.69	43.69	42.67
Feb.	44.20	50.04	50.04	20.83
Mar.	40.64	66.80	69.60	30.73
Apr.	12.7	84.84	99.31	65.28
May.	0.00	163.07	168.66	147.83
Jun.	0.00	180.09	202.95	175.26
Jul.	0.00	189.23	219.20	187.96
Aug.	0.00	172.21	197.36	171.70
Sep.	0.00	122.17	136.39	117.36
Oct.	12.45	87.12	86.11	65.03
Nov.	24.89	38.35	39.62	23.37
Dec.	30.73	43.43	43.43	43.43

**Table 3:** Monthly effective precipitation (EP) in DPWD and evapotranspiration (ET) of almond trees (adapted and modified from RMC Water and Environment, 2013).

Based on the 2017 census of agriculture (USDA-NASS, 2019), it was assumed that almonds yield and price were 246.6 gr/m<sup>2</sup> and  $6.61 \times 10^{-3}$  \$/gr on average. The starting money (i.e., financial resources that each farmer owns) of each farm was determined according to its area (\$1.63 × farm size (m<sup>2</sup>)) to make it simple. Selling their crops and paying for water supply were the only factors that increased and decreased farmers' money, respectively. Unit price of water resources were as follows (RMC Water and Environment, 2013; USDA-NASS, 2019): 1) CVP 0.0291-0.0486 \$/m<sup>2</sup>; 2) Transferred Water 0.0778-0.3210 \$/m<sup>2</sup>; 3) Groundwater 0.0778-0.1265 \$/m<sup>2</sup>; and 4) Recycled Water 0.0486-0.1459 \$/m<sup>2</sup>. The user can set CVP and recycled water unit prices, while transferred water and groundwater unit prices were randomly chosen within their range. It should be mentioned that the unit prices of CVP and recycled water remained constant during the simulations for simplification. Moreover, although transferred water and groundwater unit prices could also be dependent on various factors (e.g., climate), we decided to stochastically simulate them using the historical data from the study area for the sake of parsimony.

$$Money_{year=i+1} = Money_{year=i} + 1.63 \times \left( 1 - \left( \frac{total \ days \ without \ water_{year=1}}{365} \right) \right) \times farm \ size$$
(3)

Where farm size was the area of the farm in m<sup>2</sup> unit. It should be mentioned that in the model, the groundwater and transferred water did not have any limitations in terms of the water volume that they could provide for farmers. However, according to the NVRRWP's feasibility study, groundwater is hard to find in the Southern part of the DPWD, and if found, its quality is poor (RMC Water and Environment, 2013). Therefore, it was assumed that only 20% of Southern farmers who choose groundwater as their alternative water source would find suitable water to irrigate their farms. The remaining Southern farmers who choose groundwater as their alternative source of water (80%) would experience water shortage. Also, if farmers want to use recycled water, but the amount of recycled water is not enough to supply all their irrigation requirements, they will use groundwater to supplement their water supply.

*Water reuse adoption sub-model*: Unlike the physical sciences, where the constituents act in a known pattern (under specific conditions), human behavior changes dynamically and adaptively based on each situation. Therefore, understanding human behavior over time and space has been challenging for researchers (Crooks et al., 2018). The challenges are 1) choosing a theory that is applicable for the study, 2) making the theory formal for use, and 3) establishing the casualties in theory (Crooks et al., 2018; Schlüter et al., 2017). Researchers have used two scientific approaches to model human behavior, including artificial intelligence and conceptual cognitive approaches. The most widely used methods in modeling human decision-making in ABM can be further divided into three groups: behavioral frameworks, mathematical approaches, and conceptual cognitive models (Crooks et al., 2018).

While many theories exist, we chose to use the Theory of Planned Behavior (TPB) (Ajzen, 1991) in this study because it provides a realistic decision-making framework and is widely used to explain and predict human behavior in various disciplines (Wang et al., 2019). The TPB describes how a person's behavioral intention can determine the probability of the implemented behavior (Ajzen, 1991). This theory claims that control beliefs, subjective norms, and attitudes influence behavioral intention, thus, affect the person's decisions (equation (4)) (Ajzen, 1991).

$$I \cong a \times AT + b \times SN + c \times PBC \tag{4}$$

Where *I* refers to intention, *a*, *b*, and *c* are empirical weights for the parameters, and *AT*, *SN*, and *PBC* refer to attitude, subjective norm, and perceived behavioral control, respectively. A person's attitude toward a behavior is how the person thinks and feels about the behavior and reflects his expectations and evaluations of the behavior (Ajzen, 1991). The subjective norm describes the support given by significant others, such as family, friends, and co-workers. It can be divided into two subsections, including injunctive norms, describing whether others encourage the behavior, and descriptive norms, describing whether others do the behavior as well or not (Ajzen, 1991). The perceived behavioral control illustrates the extent to which a person feels capable and has confidence in their ability to execute the desired behavior and can overcome the barriers and challenges of implementing the desired behavior (Ajzen, 1991). TPB was utilized in various studies to simulate human decision-making dynamics and their effects on water resources (Gilg and Barr, 2006; Koutiva and Makropoulos, 2016; Pouladi et al., 2019; Yazdanpanah et al., 2014).

At the start of each year, this sub-model determines the farmers who have already adopted water reuse. Water reuse adoption means that the farmer has accepted to consider recycled water as an alternative water supply for irrigation if it is available. If it is cheaper than other options, the farmer will use it for irrigation. This sub-model simulates the dynamics of farmers' water reuse adoption using cognitive mapping. The TPB and probabilistic models inspire the farmers' cognitive map. This framework enables researchers to evaluate the effects of farmers' personal attitudes, peer influence (other farmers), financial situation, and customers' perception toward buying products irrigated with recycled water on agricultural water reuse adoption.

This sub-model uses survey data collected by Suri et al. (2019), who surveyed farmers' perceptions toward agricultural water reuse in the Southwest and Mid-Atlantic, USA. Of note is that we only used the survey data collected from Southwest farmers. Suri et al. (2019) concluded that factors including age, water availability concern, knowledge about water reuse, access to recycled water, education level, being aware of the importance of water reuse, race, and gender were significantly associated with farmers' perceptions toward agricultural water reuse adoption (Figure 3a and Tables S1 and S2 in the Supplementary Material). Based on the data, probabilities of different levels of these parameters were calculated to be used in the sub-model (Tables S2, S3, S4 and S5 in the Supplementary Material). The sub-model uses these results to simulate the dynamics of farmers' attitudes toward water reuse, where their attitude at the end of this process could be either positive or negative. Farmers are linked to each other based on the "Preferential Attachment," the social network introduced by Barabási and Albert (1999) to determine the subjective norm. Farmers with a positive attitude toward water reuse send positive messages (the number of messages is randomly chosen between 1-5) to their peers and vice versa. Farmers sum up all the positive and negative messages that they receive. If this sum is positive, their subjective norm toward water reuse would be positive, and vice versa.



Figure 3: (a) Water reuse adoption sub-model framework; (b) Wastewater treatment plants flowchart

The control belief component comprises two parts (Figure 3a). For the first part, which is financial analysis, farmers assess whether they can afford to pay for the recycled water next year. For this, based on their water shortage in the previous year, the unit price of recycled water, and their available money, farmers can determine whether recycled water is an appropriate option for them or not. The second part considers customers' perceptions based on literature as an exogenous factor. According to Fielding et al. (2019), the acceptance of products irrigated with recycled water ranges from 44% to 90% among customers. Therefore, the customers' perceptions were randomly chosen between 44% to 90%. Theoretically, the control belief component results could be positive or negative.

Finally, the behavioral intentions of the farmers are determined according to equation (4). This study assumed that these three factors equally affect farmers' decisions (a = b = c = 1). At the start of the model, farmers' behavioral intention is set to neutral. However, this sub-model determines those farmers whose behavioral

intention changes to positive or negative at the beginning of each year. Farmers' behavioral intentions change to positive only if at least the results of two components (attitude, subjective norm, and control belief) are positive; otherwise, it changes to negative.

#### 2.2.1.2 Wastewater treatment plants

Wastewater treatment plants is another group of agents in the model. These agents try to provide recycled water to farmers who select recycled water as their alternative water resource (Figure 3b). The location, volumetric rate of treated wastewater effluent, effluent water quality, and stored water volume of these agents are set at the start of the model according to each case. It was assumed that wastewater treatment plants in the WRAF model first serve farmers closer to their locations. There were two wastewater treatment plant agents in this study, named Modesto and Turlock. The user could set the average daily flow of treated wastewater effluent produced by these treatment plants. Each wastewater treatment plant first sent its effluent to storage ponds with a maximum of 6,167,400 m<sup>3</sup> storage (to store treated wastewater when there was no demand). Of note is Modesto treatment facility already has storage with a capacity of 6,167,400 m<sup>3</sup>, and the Turlock treatment facility is planning storage soon. This study assumed that the recycled water was conveyed from these storage ponds to each farm's irrigation system using pipes. It should be mentioned that five recycled water delivery alternatives were studied in the NVRRWP feasibility study (RMC Water and Environment, 2013); the second alternative included the piping system used in this study. It was assumed that the piping system could deliver all the recycled water that farmers buy each day for simplification. Moreover, wastewater treatment plants collect the money from recycled water sales to farmers based on the unit price of recycled water (set by the user). Of note is that the actual expansion plans for Modesto and Turlock wastewater treatment plants were also considered in this study (Figure 4).



**Figure 4:** Flow rates of Modesto and Turlock wastewater treatment plants years after starting the agricultural water reuse project (adapted from RMC Water and Environment, 2013). CMD = Cubic Meters per Day.

#### 2.2.2 Environment

Inspired by the AgriPoliS model (Happe et al., 2006), WRAF models the environment in a stylized manner. The environment is divided into several equally-sized cells similar to a chessboard, all of which are associated with attributes, including farm size and crop type. To input almond farms into the WRAF model, almond farms GIS maps were created using San Joaquin (San Joaquin County Geographic Information Systems, 2020), Stanislaus (Stanislaus County GIS Division, 2014), and Merced (Merced County GIS, 2019) counties land parcels and the 2018 crop data layer (U.S. Department of Agriculture, 2018) maps (Figure 1). According to NVRRWP's feasibility study (RMC Water and Environment, 2013), these farms were divided into three regions (Northern, Central, and Southern). This map was used as an input for the WRAF model, determining the farm area, region, and crop type.

## 2.3 Model verification, sensitivity analysis, and scenario experiments

**Verification:** The goal of verifying a model is to ensure that the implemented model matches its design (North and Macal, 2007; Patel et al., 2012). This was achieved through code walkthroughs and testing the input parameters to evaluate their effects on the model results. During the verification process, simulations were run for 84 years to simulate three consecutive cycles of growing almond trees in the DPWD. Several sub-model runs showed that 84 years was enough to capture the dynamics of diffusion of water reuse among the farmers.

**Sensitivity analysis:** In addition to verification, we also carried out an extensive sensitivity analysis. The main goal of sensitivity analysis is to identify independent variables that significantly influence the model's dependent variables (Happe et al., 2006). Emergent properties are inherent to agent-based models, making it difficult to easily investigate how agent-based models' assumptions and inner interactions influence the model behavior (Happe et al., 2006). Therefore, a formal sensitivity analysis was applied to the developed model to address these challenges in a structured way. To do so, the Design of Experiments (DOE) statistical techniques were used for the sensitivity analysis. These techniques enable researchers to study the details of the model's dynamics and evaluate different input parameters' influence on the output parameters. They also help study the simulation results using a common basis and help detect the potential problems in the model's logic (Happe et al., 2006; Kleijnen, 2005). DOE enables researchers to test a subset of all possible combinations of input parameters, called experimental design, reducing tests and saving time and money while effectively evaluating the effects on output parameters (Happe et al., 2006). However, it should be mentioned that the resulting metamodels of DOEs are very coarse and cannot fully capture the models' complex behavior (Happe et al., 2006).

In this study, as the screening step, seven factors were considered by the authors to assess their influence on the WRAF model (Table 4). The authors expected these factors to significantly influence the farmer's water consumption dynamics in this model. A fractional factorial design was utilized to design the screening experiments for investigating the factors' preliminary significance and their interactions (resolution: IV, fraction: 1/8). All the factors were evaluated at two levels, including low (-) and high (+) levels (Table S6 in the Supplementary Material). Four center points were also included in the design, estimating the experimental variance and checking the loss of linearity between the levels of the factors (the curvature test) (Martendal et al., 2007) (Table S6). All the other factors were kept constant during the experiments.

	Factor- unit	Description	Range
A	Modesto (m³)	The daily flow rate of the Modesto wastewater treatment plant discharge into its storage pond	55,646-189,271
В	Turlock (m³)	The daily flow rate of the Turlock wastewater treatment plant discharge into its storage pond	47,696 -189,271
С	Primary-water-source (%)	The percentage of the contracted CVP water available for farmers for their irrigation each year	0-100
D	CVP-water-price (\$/m <sup>2</sup> )	The unit price of the contracted CVP water, which farmers have to pay upon using	\$0.0291-\$0.0486
E	Water-reuse-expenses (\$/m <sup>2</sup> )	The unit price of the recycled water that farmers have to pay upon using	\$0.0486-\$0.1459
F	Transferred-water	Availability of transferred water for farmers to use for their irrigation each year	Available (-1), not-available (+1)
G	Groundwater	Availability of groundwater for farmers to use for their irrigation each year	Available (-1), not-available (+1)

Table 4: Factors considered in the screening step of the study

Farmers' total recycled water consumption was used as the response variable in this study. Sensitivity analysis simulations included 84 years, and the data were gathered at the end of each year. Some of the parts of the WRAF model were simulated stochastically, including farmers' irrigation method, location of farmers in each of the regions, unit prices of groundwater and transferred water, access to groundwater with decent quality for irrigation in the southern region, and farmers' age, sex, education, race, knowledge, access, concern, importance, attitude, and network. Therefore, all the 20 simulations (Table S6) were replicated ten times. The means of the results were used to form the meta-model, considering the stochasticity in the results. Ten

replications were sufficient for the screening stage as the results were robust against the stochastic processes of the WRAF model. This robustness was illustrated using standard error bands in the simulation results section.

The linear regression models were fitted to data to determine the effects of factors and compare them, using the data from years 27, 55, and 84. These three years were selected because they were at the end of the three cycles of almond production. The regression models (second-order polynomial) were fitted to the data using the backward elimination method (alpha = 0.05). Significant factors and their effects were identified for each year using the linear regression model, analysis of variance, and Pareto charts ( $P \le 0.05$ ).

Finally, six scenarios were defined using the factors and their effects that were found influential on the total recycled water consumption to test the screening stage results. Scenarios were designed to shed light on how the factors affect recycled water consumption dynamics, including changing between 3 levels (low, middle, and high levels) of one of the factors each time (Table 5). Other factors and parameters were fixed each time to decrease their variation effect on the response. All the simulations were replicated ten times and run for 84 years. Total recycled water consumption was evaluated during these simulation experiments as the response.

Term	Factor	Scenario 1	Scenario 2	Scenario 3	Scenario 4	Scenario 5	Scenario 6
E	Water-reuse- expenses (\$/m³)	0.05, 0.10, 0.15	0.10	0.10	0.10	0.10	0.10
G	Groundwater	A1	A, NA <sup>2</sup>	А	А	А	А
BD	Turlock (m³) × CVP-water- price (\$/m³)	118,483 × 0.04	118,483 × 0.04	47,696 × 0.03, 118,483 × 0.04, 189,271 × 0.05	118,483 × 0.04	47,696, 118,483, 189,271 × 0.04	118,483 × 0.04
A	Modesto (m³)	122,647	122,647	122,647	55,645, 122,647, 189,271	122,647	122,647
В	Turlock (m³)	118,483	118,483	47,696, 118,483, 189,271	118,483	47,696, 118,483, 189,270	118,483
С	Primary- water-source- %	50	50	50	50	50	0, 50, 100

**Table 5:** Details of simulation experiments for testing the sensitivity analysis results in this study.

<sup>1</sup> Available; <sup>2</sup> Not-available

*Simulation experiments:* After verifying that the WRAF model was working according to its design (verification) and identifying the influential factors (sensitivity analysis), the WRAF model was utilized for simulation experiments. Four scenarios were defined for the tests using the WRAF model (Table 6). The first scenario (I) was according to the NVRRWP plan. The second and third scenarios (II and III) were defined to simulate possible moderate and severe drought conditions in the DPWD, respectively. The fourth scenario (IV) was also set to simulate possible wet conditions in the DPWD.

Table 6: Specifications of	he simulation experiments	scenarios in this study.
----------------------------	---------------------------	--------------------------

Term	Factor	Scenario I (normal)	Scenario II (moderate drought)	Scenario III (severe drought)	Scenario IV (wet)
С	Primary-water-source (%)	35	20	5	100 <sup>1</sup>
D	CVP-water-price (\$/m <sup>3</sup> )	0.040	0.045	0.050	0.030
E	Water-reuse-expenses (\$/CM)	0.09	0.10	0.11	0.08
F	Transferred-water	Available	Not-available	Not-available	Available
G	Groundwater	Available	Available	Not-available	Available

<sup>1</sup> There were several years that CVP could provide 100% to the DPWD's farmers in the past (i.e., 1995, 1998, and 2006).

Accordingly, all scenarios were simulated using the NVRRWP expansion planning regarding Modesto and Turlock's effluent flow rates (Figure 4). All the other factors were set according to the climatic conditions of the scenarios. Regarding the available CVP water supply percentage, 35%, 20%, 5%, and 100% were used for the normal, moderate drought, severe drought, and wet conditions scenarios, respectively (Table 6). The unit prices of CVP water and the unit prices of recycled water provided to the DPWD's farmers were reported under scenarios I, II, III, and IV in Table 6. It was assumed that the transferred water supply was not available to farmers in moderate and severe drought scenarios (Table 6). Groundwater was also not available as an irrigation source for farmers in the severe drought scenario (Table 6). All the simulations were repeated ten times and for 84 years of almond production.

# 3. Results

#### 3.1 Verification and sensitivity analysis

Verification: The results of representative runs are depicted in Figure 5. Figures 5a and 5b illustrate the number of farmers based on their water sources in years one and 84, respectively. It was apparent that precipitation and CVP water resources were insufficient to ensure that all farmers get enough water to irrigate their almond orchards throughout the year, especially from May (day 121) through September (day 274). In this period, farmers had to supplement their irrigation with groundwater, transferred water, and/or recycled water. Results also showed that introducing recycled water in this area could decrease the number of farmers who use groundwater or transferred water for irrigation. During these months, the number of farmers with water shortages also decreased by using recycled water (Figure 5b). The volume of recycled water stored at the storage ponds is depicted in Figures 5c and 5d. As the number of farmers who use recycled water increases, the recycled water volume in these storage ponds decreases. After several years, this depletion even reaches a level that these ponds become empty at some point, as the wastewater treatment effluent could not recharge them fast enough relative to the demand. This implies that the two treatment plants could not completely satisfy the need for recycled water in the DPWD after several years since starting the agricultural water reuse project. For example, the total amounts of consumed recycled water in years two and 84 by DPWD farmers were also illustrated in Figure 5e and 5f, respectively. The results of these two figures were consistent with Figures 5a, 5b, 5c, and 5d. Comparing these two figures shows how, as time passed, the amount of recycled water supplied by the two storages was not enough to support all farmers who needed it in year 84.

Sensitivity analysis: The factors with a significant effect ( $p \le 0.05$ ) on total recycled water consumption were identified using the methodology described before (Table 7). The unit price of recycled water was the most influential factor, negatively affecting the total recycled water consumption. This clearly showed how recycled water pricing was important for this water reuse project. The higher the unit price of recycled water, the lower the total recycled water consumption. The second influential factor in the total recycled water consumption was groundwater availability. When the groundwater was available for irrigation, less recycled water consumption was the interaction of the Turlock wastewater treatment plant's effluent flow rate and the unit price of the CVP water. This result demonstrated that these two factors affected the total recycled water consumption only if they were changed simultaneously. Other factors that could significantly affect the total recycled water consumption only if they water available to the DPWD's farmers (primary-water-source). The Modesto wastewater treatment plant's effluent flow rate and the percentage of CVP water available to the DPWD's farmers (primary-water-source). The Modesto wastewater treatment plant's effluent flow rate and the percentage of CVP water available to the DPWD's farmers (primary-water-source). The Modesto wastewater treatment plant's effluent flow rate and the percentage of CVP water available to the DPWD's farmers (primary-water-source). The Modesto wastewater treatment plant's effluent flow rate positively influenced the total recycled water consumption. However, the "primary-water-source" factor negatively affected the total recycled water consumption.



**Figure 5:** Representative simulation results: farmers' water resources distribution in year one (a) and year 84 (b); available recycled water in the storage ponds of Modesto (c) and Turlock (d) wastewater treatment plants; total recycled water used by farmers in year two (e) and year 84 (f).

**Table 7:** Significant factors effects ( $p \le 0.05$ ). The responses are total recycled water consumption at the end of years 27, 55, and 84.

Term	Factor	Effect (t = 27)	Effect (t = 55)	Effect (t = 84)
E	Water-reuse-expenses	-237,846	-268,040	-274,850
G	Groundwater	203,309	233,185	235,435
BD	Turlock×CVP-water-price	194,785	218,543	231,266
А	Modesto	77,366	136,161	141,772
В	Turlock	82,166	120,055	134,656
С	Primary-water-source	-111,785	-125,721	-125,586
	$R^2_{adj}$	91.92%	93.68%	94.1%

Residual plots were used to check whether the regression model was adequate and met the analysis' assumptions (ordinary least squares assumptions) (Figure 6). The normal probability plot of the residuals approximately followed a straight line, confirming that the residuals were normally distributed (Figures 6b, 6d, and 6f). The versus fits plots of the residuals illustrated that the residuals fell randomly on both sides of 0, with no distinguishable patterns, confirming that the residuals were randomly distributed and had constant variance. As the versus order plots of the residuals demonstrated no trend or pattern, it was concluded that the residuals were not correlated. Histograms of the residuals were also used, showing some skewness in the distributions of the residuals.



**Figure 6:** Pareto charts of standardized effects and residual plots in t = 27 [(a) and (b)], t = 55 [(c) and (d)], and t = 85 [(e) and (f)] for cumulative recycled water consumption as the response variable.

#### 3.1.1 Results of testing the screening stage

The results of increasing the unit price of recycled water clearly showed how this increase could affect the water reuse project's future (Figure 7a) by decreasing the total recycled water consumption. These results are in accordance with the sensitivity analysis results presented above. The results also illustrated that the differences between total recycled water consumption increased as time went on and reached a plateau (Figure 7a). Also, the total recycled water consumption could be decreased to zero if the unit price of recycled water was not competitive with other alternative water resources (e.g., \$0.15 per m<sup>3</sup>). Therefore, decision-makers need to pay extra attention to recycled water pricing, and in some cases, state or government incentives might be required. As the water reuse projects are costly, the decision-makers may tend to sell the recycled water as expensive as possible to compensate for these projects' high costs. However, if the recycled water is not priced right, it may lead to the project's failure.



**Figure 7:** The effects of (a) E = the unit price of recycled water and (b) G = groundwater availability on the cumulative recycled water consumption (MCMY = Millions Cubic Meters per Year).

Furthermore, the results illustrated that groundwater availability significantly affected the total recycled water consumption (Table 7 and Figure 7b). Based on these results, when groundwater was not available for farmers to use, they used more recycled water than when the groundwater was available. This result confirmed that limitation on groundwater use by any means (e.g., drought or regulations) might result in increased use of recycled water by farmers for their irrigation purposes. The simulation results of the interaction of the Turlock wastewater treatment plant's effluent flow rate and the CVP water's unit price (BD), the Turlock and Modesto wastewater treatment plants' effluent flow rates (B and A) were not consistent with sensitivity analysis results (Figures 8a, 8b and 9a). The results demonstrated that these factors were not significantly influential on the total recycled water consumption. This inconsistency could be due to the fact that was mentioned before; the meta-model resulting from the screening stage of DOEs is coarse and cannot fully represent agent-based models' complex behavior (Happe et al., 2006). Simulation results confirmed the effects of available primary water sources on total recycled water consumption (Figure 9b). These results showed that their total recycled water consumption increased significantly by decreasing the availability of farmers' primary water source (CVP). This result also indicated that variations in the amount of water provided to farmers by the CVP due to climate change and local restrictions could affect the amount of recycled water farmers consumed to supplement their primary water source. Like groundwater, decision-makers must pay attention to the effects of this factor on the sustainability of agriculture and water resources in this area.



**Figure 8:** The effects of (a) BD = the interaction of Turlock effluent flowrate and the unit price of the CVP water and (b) A = the Modesto effluent flowrate on the cumulative recycled water consumption (MCMY = Millions Cubic Meters per Year).



**Figure 9:** The effects of (a) B = the Turlock effluent flow rate and (b) C = the available percentage of the CVP water on the cumulative recycled water consumption (MCMY = Millions Cubic Meters per Year).

#### 3.2 Simulation experiments results

The results illustrated that implementing agricultural water reuse in the DPWD could successfully increase the district's water sustainability. The total water shortage was decreased by 41%, 32%, 57.7%, and 32% in scenarios I, II, III, and IV, respectively. The greatest decline in total water shortage was under the third scenario (severe drought). Under this scenario, the total water shortage tumbled by 57.7% as the water reuse project started and reached a plateau in year 30. In other scenarios, the total water shortage had a gradual declining trend (Figure 10a). The results also demonstrated that agricultural water reuse in the DPWD could significantly decline farmers' total groundwater consumption (Figure 10b). In scenarios where the groundwater was available to farmers for their irrigation practices (I, II, and IV), the agricultural water reuse project reduced the total groundwater consumption by 54.7%, 38.2%, and 74.1%, respectively. The total groundwater consumption decreased gradually from 56,337,638 CMY to 25,501,312 CMY (30,836,429 CMY decrease), 66,367,901 CMY to 41,025,195 CMY (25,342,707 CMY decrease), and 37,822,253.58 CMY to 9,791,996 CMY (28,030,258 CMY decrease) over the 84 years of simulation under scenarios I, II, and IV, respectively.

The results further depicted that the total recycled water consumption had a rising trend in all scenarios (Figure 11a). The total recycled water consumption increased from zero CMY to 46,266,156 CMY, 34,734,190 CMY, 60,339,053 CMY, and 43,005,139 CMY under scenarios I, II, III, and IV over 84 years, respectively. However, the yearly recycled water consumption reached a plateau under the severe drought scenario (III), indicating that the recycled water production could not keep up with the demand after year 30. Total transferred water consumption results also showed that implementing agricultural water reuse in the DPWD could successfully decrease the transferred water consumption for irrigation by the DPWD's farmers in scenarios where it was available for use (I and IV). The transferred water consumption was decreased by 55% and 91% in these scenarios, respectively (Figure 11b). The trend of adopting recycled water by farmers was consistent with the Diffusion of Innovations Theory, which describes the pattern and speed at which new ideas, practices, or

products spread through a population (Rogers et al., 2005) (Figure 12). These results further showed that water reuse adoption was a gradual process. The trends of scenarios I, II, and IV were almost similar; however, under scenario III, the trend was significantly different from others, with a more gradual increase in the number of farmers who adopted agricultural water reuse.



Figure 10: The results of the simulation experiments: (a) total water shortage and (b) total groundwater consumption.



**Figure 11:** The results of the simulation experiments: (a) cumulative recycled water consumption and (b) cumulative transferred water consumption.



Figure 12: The results of the simulation experiments: number of farmers who adopted agricultural water reuse.

### 4. Discussion

As discussed in Section 1, the aim of this paper was to fill a gap with respect to micro-scale dynamics of agricultural water reuse adoption by farmers and its impacts on local water resources. In what follows, we will provide a discussion of the main findings of this paper before presenting model limitations and areas of further work. Through our systematic sensitivity analysis via fractional factorial experimental design, we were able to identify the most important parameters influencing the model output (i.e., total recycled water consumption). According to the literature, the financial aspects of water reuse projects are of the most important factors

affecting these projects (Asano et al., 2007; Bixio et al., 2008; Lazarova & Bahri, 2005; Sheikh et al., 2018; Shoushtarian & Negahban-Azar, 2020). Similarly, the results of our sensitivity analysis identified the unit price of recycled water as the most important parameter affecting agricultural water reuse in the model which is the same as in the real world (e.g., Sheikh et al., 2018). It should also be noted that such sensitivity analysis is not the norm with agent-based models; many studies either neglect such sensitivity analysis or conduct it unsystematically (Borgonovo et al., 2022; Saltelli et al., 2019, 2020; Utomo et al., 2018).

The model was further used to analyze various water reuse scenarios using the most influential parameters determined by the sensitivity analysis (as described in Section 2.3). Results of the scenario analysis clearly show that recycled water consumption gradually rises with an increasing number of farmers adopting it. Furthermore, the model shows that as recycled water consumption increases, it alleviates the farmers' water shortage and could decrease groundwater over-drafting and the reliance on transferred water (Section 3). Such results suggest that recycled water as a water resource whose production does not rely on climatic conditions could also benefit farmers in this agricultural water district by providing a reliable water source for their crops. This is in accordance with agricultural water reuse literature (Asano et al., 2007; Bixio et al., 2008; Jeong & Adamowski, 2016; Lazarova & Bahri, 2005; Asano et al., 2007; Miller-Robbie et al., 2017; Nazari et al., 2012). Furthermore, this result from our models aligns with another study conducted to evaluate how water reuse can increase the robustness of a water system in the Netherlands (Pronk et al., 2021). Like our study, this study demonstrated that implementing agricultural water reuse could successfully substitute the amount of water extracted from the groundwater sources for agriculture, increasing the water system's robustness to future stresses on groundwater sources.

Now turning the discussion to limitations, one could argue that all models have their limitations, and our model is no exception. For example, as noted in Section 3, one of the assumptions made in this research was to keep the unit prices of CVP and recycled water constant during the simulation. Our rationale for this simplifying assumption was that there is currently no unified structure for pricing recycled water, and there is a huge gap in the literature regarding recycled water pricing (Abdelmoula et al., 2021; Clumpner, 2016; Gonzalez-Serrano et al., 2005; Hurlimann & McKay, 2007; Seguí-Amórtegui et al., 2005). For example, water utilities in California currently use various methods for setting recycled water prices, including base and tiered rates. This price depends on various factors (e.g., the type of water utility, the type of water user, and the water requirement volume) that increases the complexities of setting a price on recycled water. It also should be mentioned that the prices set by utilities can be changed year by year due to various factors such as the economy and environmental conditions (e.g., drought and water bodies contamination) (Clumpner, 2016). Therefore, in this study, the unit prices of CVP and recycled water remained constant for the sake of parsimony. Hence, this is one of the areas for further research. This is one of the reasons why we provide the data and code for the model so others can extend the model to include such investigations if they deem it appropriate. For example, one could change these unit prices exogenously by randomly sampling from a distribution (e.g., normal distribution) or simulate unit prices by developing a sub-model considering such things as the climatic (e.g., wet or drought conditions) and social (e.g., public acceptance of these water sources) factors as independent variables.

Another simplifying assumption we made is with respect to simulating the unit prices of transferred water and groundwater stochastically (Section 2.2.1.1). However, transferred water and groundwater costs can be impacted by various factors (e.g., climate, local governmental policies, and economy). For example, in case of any drought occurring in the study area, it is likely that the drought negatively impacts neighboring areas from which water is transferred to the study area. Therefore, the unit price of transferred water might increase due to this climatic factor. Groundwater unit prices can also be affected by drought as the available groundwater for farmers would decrease. On the other hand, suitable climatic conditions with ample rain can decrease transferred water and groundwater unit prices. The economy is another factor that can potentially indirectly impact these two water resources' unit prices. For example, increases in the cost of energy sources (e.g., gasoline and electricity) can significantly affect the costs associated with pumping water from other areas (i.e., the transferred water source) and aquifers (i.e., groundwater source) to the farms. However, including these factors in the model could significantly increase the complexities of the model presented here. Therefore, we decided not to include these in this study and leave them as suitable areas for further research.

Moreover, demographic parameters included in this model included farmers' age, race, sex, education, knowledge of recycled water, concern about water availability, the importance of recycled water, and access to recycled water. The assumption of considering these parameters constant during the simulations helped us

decrease the complexities of the model. However these parameters can significantly change due to various reasons (e.g., farmers' migration and economy) during a long period. This is one of the interesting areas of further study for researchers interested in extending this study using the data and model provided. More survey studies can be done at different times to see the changes in these parameters and their effects on model results over time. Moreover, a sensitivity analysis can be conducted to investigate the most influential demographic parameters on farmers' perceptions toward agricultural water reuse and the model results. Following on from this point, a related area for further research would be to explore time-varying sensitivity analysis as such analysis could help illustrate the dynamics of parameters' importance on the model outputs. For example, Ghoreishi et al. (2021) showed how time-varying sensitivity analysis on an agent-based model led to new insights into how financial factors impact water conservation methods for crop production. However, we feel this is beyond the scope of the current paper, but it is one reason we provide the data and source code of the model available online.

## 5. Conclusion

The use of recycled water for agricultural irrigation is a potentially viable way to address water scarcity in many parts of the world. However, the complexities of the socio-hydrological systems of water reuse projects make it very difficult to successfully investigate the implementation of these systems. Furthermore, these projects' high costs of capital, operation, and maintenance add to the systems' complexity. Investigating the socio-hydrological dynamics of agricultural water reuse projects can provide insights into these complex systems. It enables the decision-makers to test various scenarios when they intend to promote water reuse and helps them select the best management practices.

In this study, an agent-based model was successfully developed and implemented for simulating the sociohydrological dynamics of agricultural water reuse projects. We tested the model using the NVRRWP case study. The agent-based model evaluated the water consumption dynamics and how the adoption of recycled water use by farmers impacts the local water resources. The model demonstrated that agricultural water reuse could decrease total water shortage, groundwater over-drafting, and transferred water consumption if planned and managed correctly. The model, similar to the literature, suggests that decision-makers should pay special attention to the price-setting of recycled water, which was identified as the most influential factor in total recycled water consumption by farmers in the model. This study also showed how possible droughts or groundwater withdrawal regulations could increase recycled water reuse by farmers. The results also depicted that under possible drought scenarios in the future, agricultural water reuse could successfully address water supply challenges by decreasing farmers' water shortage by providing a reliable water resource for irrigation purposes.

# Acknowledgment

This research was funded by the United States Department of Agriculture-National Institute of Food and Agriculture, Grant number 2016-68007-25064, which established CONSERVE: A Center of Excellence at the Nexus of Sustainable Water Reuse, Food and Health. We would also like to thank the anonymous reviewers for providing constructive reviews, which helped improve the quality of the paper.

# References

- Abdelmoula, S., Sorour, M. T., & Aly, S. A. A. (2021). Cost Analysis and Health Risk Assessment of Wastewater Reuse from Secondary and Tertiary Wastewater Treatment Plants. Sustainability 13(23), 13125. https://doi.org/10.3390/SU132313125
- Ajzen, I. (1991). The theory of planned behavior. Organizational Behavior and Human Decision Processes, 50(2), 179–211.
- Akhbari, M., & Grigg, N. S. (2013). A framework for an agent-based model to manage water resources conflicts. Water Resources Management, 27(11), 4039–4052.
- Ali, A. M., Shafiee, M. E., & Berglund, E. Z. (2017). Agent-based modeling to simulate the dynamics of urban water supply: Climate, population growth, and water shortages. Sustainable Cities and Society, 28, 420–434.
- Asano, T., Burton, F. L., Leverenz, H. L., Tsuchihashi, R., & Tchobanoglous, G. (2007). Water reuse: issues, technologies, and applications. New York, US: McGraw-Hill.

Barabási, A.-L., & Albert, R. (1999). Emergence of scaling in random networks. Science, 286(5439), 509–512.

- Bert, F., North, M., Rovere, S., Tatara, E., Macal, C., & Podestá, G. (2015). Simulating agricultural land rental markets by combining agent-based models with traditional economics concepts: The case of the Argentine Pampas. Environmental Modelling & Software, 71, 97–110.
- Bithell, M., & Brasington, J. (2009). Coupling agent-based models of subsistence farming with individual-based forest models and dynamic models of water distribution. Environmental Modelling & Software, 24(2), 173–190.
- Bixio, D., Thoeye, C., Wintgens, T., Ravazzini, A., Miska, V., Muston, M., Chikurel, H., Aharoni, A., Joksimovic, D., & Melin, T. (2008). Water reclamation and reuse: implementation and management issues. Desalination, 218(1–3), 13–23.
- Borgonovo, E., Pangallo, M., Rivkin, J., Rizzo, L., & Siggelkow, N. (2022). Sensitivity analysis of agent-based models: a new protocol. Computational and Mathematical Organization Theory, 28, pages 52–94. https://doi.org/10.1007/s10588-021-09358-5
- Clumpner, G. (2016). Recycled Water Pricing Alternatives NBS. Last accessed on 9/28/2022 at https://www.nbsgov.com/featured/recycled-water-pricing-alternatives/
- Crooks, A. T., & Castle, C. J. E. (2012). The Integration of Agent-Based Modelling and Geographical Information for Geospatial Simulation. In Agent-Based Models of Geographical Systems. Editors: Heppenstall, A., Crooks, A. T., See, L. M., Batty, M., Springer. London. https://doi.org/10.1007/978-90-481-8927-4
- Crooks, A., Malleson, N., Manley, E., & Heppenstall, A. (2018). Agent-based modelling and geographical information systems: a practical primer. SAGE Publications Limited.
- Elsawah, S., Guillaume, J. H. A., Filatova, T., Rook, J., & Jakeman, A. J. (2015). A methodology for eliciting, representing, and analysing stakeholder knowledge for decision making on complex socio-ecological systems: from cognitive maps to agent-based models. Journal of Environmental Management, 151, 500–516.
- Eslamian, S. (Ed.). (2016). Urban Water Reuse Handbook. 1st Edition. CRC Press. Boca Raton, FL.
- FAO (2015). Irrigation potential in Africa: A basin approach. Chapter 5: Irrigation water requirements. Natural Resources Management and Environment Department. Last accessed on 9/28/2022 at http://www.fao.org/3/w4347e/w4347e0c.htm
- Farhadi, S., Nikoo, M. R., Rakhshandehroo, G. R., Akhbari, M., & Alizadeh, M. R. (2016). An agent-based-nash modeling framework for sustainable groundwater management: A case study. Agricultural Water Management, 177, 348–358.
- Fielding, K. S., Dolnicar, S., & Schultz, T. (2019). Public acceptance of recycled water. International Journal of Water Resources Development, 35(4), 551–586. https://doi.org/10.1080/07900627.2017.1419125
- Filatova, T., Verburg, P. H., Parker, D. C., & Stannard, C. A. (2013). Spatial agent-based models for socio-ecological systems: Challenges and prospects. Environmental Modelling & Software, 45, 1–7.
- Ghoreishi, M., Sheikholeslami, R., Elshorbagy, A., Razavi, S., & Belcher, K. (2021). Peering into agricultural rebound phenomenon using a global sensitivity analysis approach. Journal of Hydrology, 602, 126739. https://doi.org/10.1016/J.JHYDROL.2021.126739
- Gilg, A., & Barr, S. (2006). Behavioural attitudes towards water saving? Evidence from a study of environmental actions. Ecological Economics, 57(3), 400–414.
- Gonzalez-Serrano, E., Rodriguez-Mirasol, J., Cordero, T., Koussis, A. D., Rodriguez, J. J., & Gonzalez-Serrano, E. (2005). Cost of reclaimed municipal wastewater for applications in seasonally stressed semi-arid regions. International Water Association, 54(6), 355–269.
- Grimm, V., Berger, U., Bastiansen, F., Eliassen, S., Ginot, V., Giske, J., Goss-Custard, J., Grand, T., Heinz, S. K., Huse, G., Huth, A., Jepsen, J. U., Jørgensen, C., Mooij, W. M., Müller, B., Pe'er, G., Piou, C., Railsback, S. F., Robbins, A. M., Robbins, M., Rossmanith, E., Ruger, N., Strand, E., Souissi, S., Stillman, R.A., Vabo, R., Visser, U., DeAngelis, D. L. (2006). A standard protocol for describing individual-based and agent-based models. Ecological Modelling, 198(1–2), 115–126. https://doi.org/10.1016/j.ecolmodel.2006.04.023
- Groeneveld, J., Müller, B., Buchmann, C. M., Dressler, G., Guo, C., Hase, N., Hoffmann, F., John, F., Klassert, C., & Lauf, T. (2017). Theoretical foundations of human decision-making in agent-based land use models–A review. Environmental Modelling & Software, 87, 39–48.
- Happe, K., Kellermann, K., & Balmann, A. (2006). Agent-based Analysis of Agricultural Policies: an Illustration of the Agricultural Policy Simulator AgriPolis, its Adaptation and Behavior. Ecology and Society, 11(1): 49.
- Huber, R., Bakker, M., Balmann, A., Berger, T., Bithell, M., Brown, C., Grêt-Regamey, A., Xiong, H., Le, Q. B., & Mack, G. (2018). Representation of decision-making in European agricultural agent-based models. Agricultural Systems, 167, 143– 160.
- Hurlimann, A., & McKay, J. (2007). Urban Australians using recycled water for domestic non-potable use—An evaluation of the attributes price, saltiness, colour and odour using conjoint analysis. Journal of Environmental Management, 83(1), 93–104. https://doi.org/10.1016/J.JENVMAN.2006.02.008
- Janssen, M. A., & Ostrom, E. (2006). Empirically based, agent-based models. Ecology and Society, 11(2): 37.
- Jeong, H., & Adamowski, J. (2016). A system dynamics based socio-hydrological model for agricultural wastewater reuse at the watershed scale. Agricultural Water Management, 171, 89–107.
- Kandiah, V. K., Berglund, E. Z., & Binder, A. R. (2019a). An Agent-based Modeling Approach to Project Adoption of Water Reuse and Evaluate Expansion Plans within a Sociotechnical Water Infrastructure System. Sustainable Cities and Society. 46, 101412. https://doi.org/10.1016/J.SCS.2018.12.040

- Kandiah, V. K., Berglund, E. Z., & Binder, A. R. (2019b). An agent-based modeling approach to project adoption of water reuse and evaluate expansion plans within a sociotechnical water infrastructure system. Sustainable Cities and Society, 46, 101412.
- Kelly, R. A., Jakeman, A. J., Barreteau, O., Borsuk, M. E., ElSawah, S., Hamilton, S. H., Henriksen, H. J., Kuikka, S., Maier, H. R., & Rizzoli, A. E. (2013). Selecting among five common modelling approaches for integrated environmental assessment and management. Environmental Modelling & Software, 47, 159–181.
- Kleijnen, J. P. C. (2005). An overview of the design and analysis of simulation experiments for sensitivity analysis. European Journal of Operational Research, 164(2), 287–300. https://doi.org/10.1016/j.ejor.2004.02.005
- Kock, B. E. (2008). Agent-based models of socio-hydrological systems for exploring the institutional dynamics of water resources conflict. Massachusetts Institute of Technology.
- Koutiva, I., & Makropoulos, C. (2016). Modelling domestic water demand: An agent based approach. Environmental Modelling & Software, 79, 35–54.
- Lazarova, V., & Bahri, A. (2005). Water Reuse for Irrigation Agriculture, Landscapes, and Turf Grass. CRC Press. Boca Raton, FL.
- Martendal, E., Budziak, D., & Carasek, E. (2007). Application of fractional factorial experimental and Box-Behnken designs for optimization of single-drop microextraction of 2, 4, 6-trichloroanisole and 2, 4, 6-tribromoanisole from wine samples. Journal of Chromatography A, 1148(2), 131–136.
- Mendelsohn, R., & Dinar, A. (2003). Climate, water, and agriculture. Land Economics, 79(3), 328–341.
- Merced County GIS. (2019). Updated Merced County Assessor Parcel. Last accessed on 10/11/2022 at http://geostackmercedcounty.opendata.arcgis.com/datasets/updated-merced-county-assessor-parcel?geometry=-120.688%2C37.181%2C-120.606%2C37.193
- Miller-Robbie, L., Ramaswami, A., & Amerasinghe, P. (2017). Wastewater treatment and reuse in urban agriculture: Exploring the food, energy, water, and health nexus in Hyderabad, India. Environmental Research Letters, 12(7), 75005.
- Nazari, R., Eslamian, S., & Khanbilvardi, R. (2012). Water reuse and sustainability. Voudouris, K. (Ed) Ecological Water Quality– Water Treatment and Reuse, Intechopen, 241–254.
- Nikolic, V. V, & Simonovic, S. P. (2015). Multi-method Modeling Framework for Support of Integrated Water Resources Management. Environmental Processes, 2, 461–483. https://doi.org/10.1007/s40710-015-0082-6
- North, M. J., & Macal, C. M. (2007). Managing Business Complexity: Discovering Strategic Solutions with Agent-Based Modelling and Simulation. Oxford University Press.
- Patel, A., Crooks, A., & Koizumi, N. (2012). Simulation: An agent-based modelling approach to slum formations. Journal of Artificial Societies and Social Simulation
- , 15(4). https://doi.org/10.18564/jasss.2045
- Paul, M., Negahban-Azar, M., Shirmohammadi, A., & Montas, H. (2020). Assessment of agricultural land suitability for irrigation with reclaimed water using geospatial multi-criteria decision analysis. Agricultural Water Management. 231: 105987. https://doi.org/10.1016/j.agwat.2019.105987
- Pope, A. J., & Gimblett, R. (2015). Linking Bayesian and agent-based models to simulate complex social-ecological systems in semi-arid regions. Frontiers in Environmental Science, 3, 55.
- Pouladi, P., Afshar, A., Afshar, M. H., Molajou, A., & Farahmand, H. (2019). Agent-based socio-hydrological modeling for restoration of Urmia Lake: Application of theory of planned behavior. Journal of Hydrology, 576, 736–748. https://doi.org/10.1016/j.jhydrol.2019.06.080
- Pronk, G. J., Stofberg, S. F., van Dooren, T., Dingemans, M. M. L., Frijns, J., Koeman-Stein, N. E., Smeets, P., & Bartholomeus, R. P. (2021). Increasing Water System Robustness in the Netherlands: Potential of Cross-Sectoral Water Reuse. Water Resources Management, 35(11), 3721–3735.
- Rasoulkhani, K., Logasa, B., Presa Reyes, M., & Mostafavi, A. (2018). Understanding fundamental phenomena affecting the water conservation technology adoption of residential consumers using agent-based modeling. Water, 10(8), 993.
- RMC Water and Environment. (2013). North Valley Regional Recycled Water Program Feasibility Study (Vol. 1, Issue 1). U.S. Bureau of Reclamation.
- Rogers, E. M., Medina, U. E., Rivera, M. A., & Wiley, C. J. (2005). Complex Adaptive Systems and The Diffusion of Innovations. The Innovation Journal: The Public Sector Innovation Journal, 10(3), 1–26.
- Saltelli, A., Aleksankina, K., Becker, W., Fennell, P., Ferretti, F., Holst, N., Li, S., & Wu, Q. (2019). Why so many published sensitivity analyses are false: A systematic review of sensitivity analysis practices. Environmental Modelling & Software, 114, 29–39. https://doi.org/10.1016/J.ENVSOFT.2019.01.012
- Saltelli, A., Bammer, G., Bruno, I., Charters, E., di Fiore, M., Didier, E., Nelson Espeland, W., Kay, J., Io Piano, S., Mayo, D., Pielke, R., Portaluri, T., Porter, T. M., Puy, A., Rafols, I., Ravetz, J. R., Reinert, E., Sarewitz, D., Stark, P. B., Stirling, A., Van der Sluijs, J., Vineis, P. (2020). Five ways to ensure that models serve society: a manifesto. Nature 582(7813), 482–484. https://doi.org/10.1038/d41586-020-01812-9
- San Joaquin County Geographic Information Systems. (2020). San Joaquin County GIS Map Server. Last accessed on 10/11/2022 at https://www.sjmap.org/GISDataDownload.htm
- Schlüter, M., Baeza, A., Dressler, G., Frank, K., Groeneveld, J., Jager, W., Janssen, M. A., McAllister, R. R. J., Müller, B., Orach, K., Schwarz, N., & Wijermans, N. (2017). A framework for mapping and comparing behavioural theories in models of social-ecological systems. Ecological Economics, 131, 21–35. https://doi.org/10.1016/j.ecolecon.2016.08.008

- Seguí-Amórtegui, L., Alfranca-Burriel, O., Guerrero-Garcia-Rojas, H., & Moeller-Chávez, G. (2005). Price for water reclamation, how can it be established? A Mexican case study. Water Supply, 5(2), 87–96. https://doi.org/10.2166/WS.2005.0025
- Sheikh, B., Nelson, K., Thebo, A., Haddad, B., Gardner, T., Kelly, J., Adin, A., Tsuchihashi, R., Spurlock, S., & Funamizu, N. (2018). Agricultural Use of Recycled Water: Impediments and Incentives. Water Research Foundation.
- Shoushtarian, F., & Negahban-Azar, M. (2020). Worldwide Regulations and Guidelines for Agricultural Water Reuse: A Critical Review. Water, 12(4), 971. https://doi.org/10.3390/w12040971
- Stanislaus County GIS Division (2014). Assessor's Parcels, Stanislaus County, California, 2019. Last accessed on 10/11/2022 at https://purl.stanford.edu/sq899tr7228
- Suri, M. R., Dery, J. L., Pérodin, J., Brassill, N., He, X., Ammons, S., Gerdes, M. E., Rock, C., & Goldstein, R. E. R. (2019). U.S. farmers' opinions on the use of nontraditional water sources for agricultural activities. Environmental Research, 172, 345–357. https://doi.org/10.1016/j.envres.2019.02.035
- Tillman, T., Larsen, T. A., Pahl-Wostl, C., & Gujer, W. (2001). Interaction analysis of stakeholders in water supply systems. Water Science and Technology, 43(5), 319–326.
- U.S. Department of Agriculture. (2018). CropScape NASS CDL Program. National Agricultural Statistics Service. Last accessed on 10/11/2022 at https://nassgeodata.gmu.edu/CropScape/
- USDA-NASS. (2019). 2018 Irrigation and Water Management Survey. United States Department of Agriculture. Census of Agriculture. 3(November), 1–269.
- Utomo, D. S., Onggo, B. S., & Eldridge, S. (2018). Applications of agent-based modelling and simulation in the agri-food supply chains. European Journal of Operational Research, 269(3), 794–805. https://doi.org/10.1016/J.EJOR.2017.10.041
- Wang, Y., Liang, J., Yang, J., Ma, X., Li, X., Wu, J., Yang, G., Ren, G., & Feng, Y. (2019). Analysis of the environmental behavior of farmers for non-point source pollution control and management: An integration of the theory of planned behavior and the protection motivation theory. Journal of Environmental Management, 237, 15–23.
- Wilensky, U. (1999). NetLogo. Center for Connected Learning and Computer-Based Modelling, Northwestern University, Evanston, IL. Last accessed on 10/11/2022 at http://ccl.northwestern.edu/netlogo/
- Wise, S., & Crooks, A. T. (2012). Agent-based modelling for community resource management: Acequia-based agriculture. Computers, Environment and Urban Systems, 36(6), 562–572.
- Yazdanpanah, M., Hayati, D., Hochrainer-Stigler, S., & Zamani, G. H. (2014). Understanding farmers' intention and behavior regarding water conservation in the Middle-East and North Africa: A case study in Iran. Journal of Environmental Management, 135, 63–72.